Author’s response to reviews

Title: Where do students in the health professions want to work?

Authors:

Deborah Schofield (deborah.schofield@ncahs.health.nsw.gov.au)
Susan Fletcher (deborah.schofield@ncahs.health.nsw.gov.au)
Jeffery Fuller (jeffrey.fuller@ncahs.health.nsw.gov.au)
Hudson Birden (hudson.birden@ncahs.health.nsw.gov.au)
Sue Page (sue.page@ncahs.health.nsw.gov.au)

Version: 5 Date: 6 July 2009

Author’s response to reviews: see over
Dr Deborah Schofield,
Associate Professor and Director of Research,
Northern Rivers University Department of Rural Health,
School of Public Health, Faculty of Medicine, University of Sydney.
PO Box 3074, Lismore, NSW, 2480

06 July 2009

Dr Mario Dal Poz
Editor-in-Chief
Human Resources for Health
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB, UK

Dear Dr Dal Poz

Re: Revision of “Where do students from different disciplines want to work?”

We have made the following revisions to the paper “Where do students from different disciplines want to work?”:

- Updated main titles in word document to match that in the submission system,
- Altered the graphics in figure 4 to make sure it is readable,
- Altered the figure citation so they are consistent,
- Updated the abstract in the manuscript to match that in the submission system,
- Repositioned the tables in the manuscript to after the references, and
- Altered the affiliations so that the address is only written once.

Best regards

Deborah Schofield